Author’s response to reviews

Title: Case report: two novel VPS13B mutations in a Chinese family with Cohen syndrome and hyperlinear palms

Authors:
Sha Zhao (zhaosha666@163.com)
Zhenqing Luo (2938610199@qq.com)
Xiao Zhenghui (xiaozh888@126.com)
Liping Li (13974871993@163.com)
Rui Zhao (ruizhao@aliyun.com)
Yongjia Yang (yongjia727@aliyun.com)
Yan Zhong (zhongyan@163.com)

Version: 4 Date: 08 Nov 2019

Author’s response to reviews:

1. Acknowledgments

Please move the statement “Gratitude is extended to the subject family for their cooperation and participation.” from underneath the ‘Declarations’ header to an ‘Acknowledgements’ sub-section in the ‘Declarations’ section of your manuscript.

Please note that the Declarations section encompasses the sub-sections: Ethics approval and consent to participate, Consent for publication, Availability of data and material, Competing interests, Funding, Authors' contributions, Acknowledgements.

ans: We have moved the statement “Gratitude is extended to the subject family for their cooperation and participation.” from underneath the ‘Declarations’ header to an ‘Acknowledgements’ sub-section in the ‘Declarations’ section.

2. Authors’ contributions

a) In the section “Authors’ contributions”, please provide further details regarding the individual contributions of Liping Li.
ans: The author LPL is added to the contribution of ‘designed the study and revised the manuscript’ according to the fact that she is participated in this part.

b) Please amend the initials “ZHX” to “XZH”.
ans: This author is ZHX (her family name is Xiao, her first name is Zhenghui).
c) Please amend the initials “LLP” to “LPL”.

ans: We have changed this point according to the Editor’s suggestion.